Author's response to reviews

Title: Implementation of a workplace smoking ban in bars: The limits of local discretion

Authors:

Theresa Montini (theresa.montini@nyu.edu)
Lisa A Bero (berol@pharmacy.ucsf.edu)

Version: 6 Date: 10 September 2008

Author's response to reviews:

Response to latest request for revisions:

Tables must be created using a tables tool. Tables must be divided into rows and columns and be orientated in a portrait page layout, without visible vertical lines.

RESPONSE: Tables have been recreated using the tables tool.

---------

Manuscript sections - Manuscript sections should include (in the following order): Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors' contributions; Acknowledgements; References; Figure legends (if any); Tables (if any); Description of Additional files (if any).

RESPONSE: Appendices have been submitted as “additional files.”

---------

Endnote - There is a discrepancy between the number of references in your manuscript reference list, and the number extracted by our system. This can be seen on the upload page of the submission system in the main manuscript box. For help please visit: http://www.biomedcentral.com/info/ifora/endnote/#technical help

RESPONSE: Checked and rechecked. Some references are cited more than once in different places in the manuscript. Some citations contain more than one reference – but all is correct. Do not understand the discrepancy – please give more details as to what the problem is.